Author’s response to reviews

Title: Multiple Intracranial Arteries Stenosis Influence the Long-term Prognosis of Symptomatic Middle Cerebral Artery Occlusion

Authors:

Lili tian (ltian28@gmail.com)
Xuanye Yue (yuexy2002@hotmail.com)
Gangming Xi (xgmsys@163.com)
Youmeng Wang (fyswym@163.com)
Zongyou Li (fyszy1@163.com)
Ying Zhou (fysyyzy@163.com)
Gelin Xu (xfliu2@vip.163.com)
Xinfeng Liu (xfliu2@vip.163.com)

Version: 2 Date: 11 November 2014

Author’s response to reviews:

Dear Nhell

We are so appreciated for your letter on our manuscript (MS ID: 2002759651148403). We are truly grateful to your suggestions on improving the manuscript, which indeed help us greatly to improve the quality of our paper. Based on your suggestions, we have made careful modifications on the original manuscript.

1 The line number has been added before each line in our manuscript.

2 Authors’ Contributions section had been placed before the Conflict of interest section.

We wish the revised edition with meet the standard of the Journal of BMC Neurology.

Best Wishes

Tian